DARSHI BANAN

3610 164th Place SE Bellevue, WA 98008 TEL: +1-425-829-2535 mail-to:jobs@darshi.1.banan.byname.net http://darshi.1.banan.byname.net http://darshi.1.banan.byname.net/Resume

Education University of Washington Seattle, WA Biology, 2007–present

1. BIOL425,426 Plant Physiology and Lab 2. BIOL440 Mycology 3. BIOL416 Molecular Basis of Plant Genetics 4. BIOL317 Plant Identification

Experience WuSu School. Bellevue, WA Owner/President, 2007 –present

WuShu teaching.

Languages My first language is English. I am conversational in French and Mandarin Chinese. I speak some Persian/Farsi.

Web Info This Resume and additional information about me is availble as an autonomous libre web site at: http://darshi.1.banan.byname.net

References Available upon request.